Author's response to reviews

Title: Prognostic significance of MCM2, Ki-67 and gelsolin in non-small cell lung cancer

Authors:

Jun Yang (Jun.Yang@RoswellPark.org)
Nithya Ramnath (Nithya.Ramnath@RoswellPark.org)
Kirsten B Moysich (Kirsten.Moysich@RoswellPark.org)
Harold L Asch (asch1@adelphia.net)
Helen Swede (helen.swede@po.state.ct.us)
Sadir J Alrawi (Sadir.Alrawi@RoswellPark.org)
Joel Huberman (Joel.Huberman@RoswellPark.org)
Joseph Geradts (joseph.geradts@duke.edu)
John SJ Brooks (john.brooks@uphs.upenn.edu)
Dongfeng Tan (Dongfeng.Tan@uth.tmc.edu)

Version: 5 Date: 29 July 2006

Author's response to reviews: see over
July 28th, 2006

Editors
BMC Cancer

RE: 3rd revised manuscript # 9873564010059150 entitled: “Prognostic significance of MCM2, Ki-67 and gelsolin in non-small cell lung cancer”

Dear Editors and Reviewers:

Based on the suggestions of editors and reviewers, we made the following point-by-point revisions:

1. Added reference for time-dependent Cox regression: Page 10, Line 6
2. Added Conclusions: Page 15, Line 1
3. Removed vertical elements (i.e., vertical lines) in all three Tables
4. Removed Figure titles from Figure files
5. Cropped Figures to minimize blank margins

We also have double-checked the format of our manuscript to ensure that it conforms to the format requirement of BMC Cancer.

Again, we appreciate the diligent work and the insightful comments of all BMC Cancer editors and reviewers.

Sincerely,

Jun Yang, M.D., Ph.D.
Department of Epidemiology
Roswell Park Cancer Institute
Elm and Carlton Streets
Buffalo, New York 14263
Phone: 001-716-845-8937
Fax: 001-716-845-8125; 716-845-8487
E-mail: Jun.Yang@RoswellPark.org